Dear Editor,

Thank you very much for your decision letter and advice on our manuscript (Manuscript #70076) entitled “Pediatric living donor liver transplantation using liver allograft after ex vivo backtable resection of hemangioma”. We also thank the reviewers for the constructive comments and suggestions. We have revised the manuscript accordingly, and all amendments are indicated by red font in the revised manuscript. In addition, our point-by-point responses to the comments are listed below this letter.

This revised manuscript has been edited and proofread by Medjaden Inc..

We hope that our revised manuscript is now acceptable for publication in your journal and look forward to hearing from you soon.

With best wishes,
Yours sincerely,

Xuan Chen
Replies to Science editor

(1) The “Author Contributions” section is missing. Please provide the author contributions;
Response: We have provided the author contributions.

(2) The authors did not provide the approved grant application form(s). Please upload the approved grant application form(s) or funding agency copy of any approval document(s);
Response: We have provided the approved grant application forms.

(3) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor;
Response: We have provided the original figure documents and arranged the figures using PowerPoint to ensure that all the graphs, arrows and text portions can be reprocessed.

(4) The “Case Presentation” section was not written according to the Guidelines for Manuscript Preparation. Please re-write the “Case Presentation” section, and add the “FINAL DIAGNOSIS”, “TREATMENT”, and “OUTCOME AND FOLLOW-UP” sections to the main text, according to the Guidelines and Requirements for Manuscript Revision;
Response: The “CASE PRESENTATION” section has been re-written according to the Guidelines for Manuscript Preparation. We have added the “FINAL DIAGNOSIS”, “TREATMENT”, and “OUTCOME AND FOLLOW-UP” sections to the main text, according to the Guidelines and Requirements for Manuscript Revision.

(5) It is unacceptable to have more than 3 references from the same journal. To resolve this issue and move forward in the peer-review/publication process, the authors must revise the reference list accordingly.
Response: We have revised the reference list so that there are no more than 3 references from the same journal in the manuscript.

Replies to company editor-in-chief

Before final acceptance, uniform presentation should be used for figures showing the same or similar contents; for example, “Figure 1Pathological changes of atrophic gastritis after treatment. A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...”. Please provide decomposable Figures (in which all components are movable and editable), organize them into a single PowerPoint file.

Response: 1. Uniform presentation has been used for figures showing the same or similar contents. 2. We have provided the original figure documents and arranged the
figures using PowerPoint to ensure that all the graphs, arrows and text portions can be reprocessed.